Author's response to reviews

Title: Ureaplasma urealyticum, Ureaplasma parvum, Mycoplasma hominis and Mycoplasma genitalium infections and semen quality of infertile men

Authors:

- Radhouane Gdoura (gdourar@yahoo.com)
- Wiem Kchaou (wiemk-chaou@yahoo.com)
- Chiraz Chaari (chiraz-chaari@yahoo.fr)
- Abir Znazen (abirznazen2001@yahoo.fr)
- Leila Keskes (leilakeskes@yahoo.fr)
- Tarek Rebai (tarek.rebai@fmsf.rnu.tn)
- Adnane Hammami (adnane.hammami@rns.tn)

Version: 5 Date: 3 November 2007

Author's response to reviews: see over
Dear Editor,

I am very grateful to you for acceptance of our manuscript entitled "Ureaplasma urealyticum, Ureaplasma parvum, Mycoplasma hominis and Mycoplasma genitalium infections and semen quality of infertile men” MS: 8064124521424744

Would you please find enclosed the following changes to the format of our manuscript. We have listed below all changes to the manuscript according to your recommendations.

* Line numbers were removed from the text.
* The full stop was removed from the end of the title.
* Unnecessary capitals were removed: The word term “AND” was replaced by “and”.
* Author surnames were decapitalised.
* Address was deleted from the affiliations.
* The sentence “All authors readed and approved the final manuscript” was added in Authors’ contributions section page 18.
* “Table2” in page 11 and 23 was changed by “Table 2”.
* All references were updated in the order they appear in the text: “The order of reference 18, 19 and 20 were checked and corrected”.

I will be ready for any suggestion or further modification that you need me to make.

Looking forward to having my manuscript published in your Journal, I would like to send you my highest respect.

Sincerely yours

GDOURA Radhouane